Author's response to reviews

Title: Evaluation of the impact of transient interruption of antiangiogenic treatment using ultrasound-based techniques in a murine model of hepatocellular carcinoma

Authors:

Sara Marinelli (sara_marinelli@libero.it)
Veronica Salvatore (veronica.salvatore@unibo.it)
Marco Baron Toaldo (marco.barontoaldo@unibo.it)
Maddalena Milazzo (maddimilazzo@hotmail.it)
Luca Croci (luca.croci5822@gmail.com)
Laura Venerandi (laura.venerandi@gmail.com)
Anna Pecorelli (pecorelli.anna@gmail.com)
Chiara Palamà ( chiarapal@live.it)
Alessia Diana (alessia.diana@unibo.it)
Luigi Bolondi (luigi.bolondi@unibo.it)
Fabio Piscaglia ( fabio.piscaglia@unibo.it)

Version: 3
Date: 15 May 2014

Author's response to reviews:

Manuscript has been changed accordingly to the latest suggestions. In particular, the term "aim" has been removed from the abstract, the name of the animal welfare committee has been specified and acknowledgments section has been moved into the correct place.